Author's response to reviews

Title: Effectiveness of Intravenous lidocaine versus Intravenous morphine for patients with Renal Colic in the Emergency Department

Authors:

Hassan Soleimanpour (h_mofid1357@yahoo.com)
Kamaleddin Hassanzadeh (drhassanzadehka@yahoo.com)
Hassan Vaezi (drhassanvaezi@yahoo.com)
Samad EJ Golzari (doctor_golzari@yahoo.com)
Maryam Soleimanpour (maryamsoleimanpour@gmail.com)
Robab Mehdizadeh Esfanjani (mehdizadeh18@yahoo.com)

Version: 2 Date: 17 December 2011

Author's response to reviews: see over
Cover letter

Dear Editor

Thank you for your thoughtful review of our manuscript. I take your concerns seriously and have addressed them to the best our abilities. I asked from one of our english literature academic staff to edit the manuscript. As we mentioned in before our abstract entitled "Effectiveness of Intravenous Lidocaine versus Intravenous morphine for patients with Renal Colic in the Emergency Department” was accepted in The 6th European Congress on Emergency Medicine (EuSEM 2010) Stockholm 22-24 October 2010, as poster presentation. This study is innovation in renal colic treatment and can be considered as an alternative method in this situation. Furthermore this study shows that this method might consider as a new and simple way to control renal colic in the emergency department. We declare that we have no competing interests. We will be glad to publish full text of it in your journal

Best regards
Authors